Author’s response to reviews

Title: Feasibility, user experiences, and preliminary effect of Conversation Cards for Adolescents© on collaborative goal-setting and behavior change: Protocol for a pilot randomized controlled trial

Authors:
Maryam Kebbe (kebbe@ualberta.ca)
Anna Farmer (anna.farmer@ualberta.ca)
Michele Dyson (mdyson@ualberta.ca)
Shannon Scott (shannon.scott@ualberta.ca)
Tara-Leigh McHugh (tmchugh@ualberta.ca)
Scot Lappa (scot.lappa@albertahealthservices.ca)
Hasu Rajani (hrajani@ualberta.ca)
Tehseen Ladha (tladha@gmail.com)
Bonnie Islam (bislam@ualberta.ca)
Lynn Jacoby (jacyjb@ualberta.ca)
Fareeha Nasir (fareeha1@ualberta.ca)
Kiran Talwar (ktalwar@ualberta.ca)
James Wincott (james.wincott@albertahealthservices.ca)
Mona Zhang (mona2@ualberta.ca)
Geoff Ball (gdball@ualberta.ca)

Version: 2 Date: 19 Nov 2019

Author’s response to reviews:

Please see Cover Letter (attached).